Author's response to reviews

Title: Antibacterial, antifungal and antioxidant activity of Olea africana against pathogenic yeast and nosocomial pathogens

Authors:

Peter Masoko (Peter.Masoko@ul.ac.za)
David M Makgapeetja (polokwanedavid@gmail.com)

Version: 5 Date: 4 November 2015

Author's response to reviews: see over
To: Editor-in-Chief, BMC Complementary and Alternative Medicine

From: Prof. P. Masoko
Department of Biochemistry, Microbiology & Biotechnology, University of Limpopo

Re: Corrections from the reviewers

The following correction from the reviewers were attend to:

Reviewer1 (Jean-de-Dieu TAMOKOU)

The reviewer is happy with the corrections made and no additional comments were send.

Reviewer2 (Thankappan Sarasam Reiniemon)

1. The author should scientifically discuss and compare the results of antibacterial and antifungal
   …..Additional paragraph was added, adding the required discussion and comparison required.
2. Add antifungal to conclusion: The word antifungal is added.
3. Language corrections: The manuscript was given to senior colleague for proofreading. Language corrected

With kind regards

Prof P. Masoko, Ph.D (Principal Author)